#### THIS CIRCULAR IS IMPORTANT AND REQUIRES YOUR IMMEDIATE ATTENTION

If you are in any doubt as to any aspect of this circular or as to the action to be taken, you should consult your licensed securities dealer, bank manager, solicitor, professional accountant or other professional adviser.

If you have sold or transferred all your Shares, you should at once hand this circular and the accompanying proxy form to the purchaser(s) or the transferee(s) or to the bank, licensed securities dealer, registered institution in securities or other agent through whom the sale or transfer was effected for transmission to the purchaser(s) or transferee(s).

Hong Kong Exchanges and Clearing Limited and The Stock Exchange of Hong Kong Limited take no responsibility for the contents of this circular, make no representation as to its accuracy or completeness and expressly disclaim any liability whatsoever for any loss howsoever arising from or in reliance upon the whole or any part of the contents of this circular.



### LEGENDARY EDUCATION GROUP LIMITED

### 傳承教育集團有限公司

(formerly known as Legendary Group Limited 創天傳承集團有限公司) (incorporated in the Cayman Islands with limited liability) (Stock Code: 8195)

### ADOPTION OF THE 2023 SHARE OPTION SCHEME; TERMINATION OF THE EXISTING SHARE OPTION SCHEME; AND NOTICE OF EXTRAORDINARY GENERAL MEETING

A notice convening the extraordinary general meeting ("EGM") of Legendary Education Group Limited (the "Company") to be held on Friday, 1 September 2023 at 11:00 a.m. at 5/F, World Interests Building, 8 Tsun Yip Lane, Kwun Tong, Kowloon, Hong Kong is set out on pages EGM-1 to EGM-3 of this circular. A proxy form for use at the EGM is enclosed.

Whether or not you intend to attend the EGM in person, you are requested to complete the enclosed proxy form in accordance with the instructions printed thereon and return the same to the branch share registrar and transfer office of the Company in Hong Kong, Union Registrars Limited at Suites 3301-04, 33/F., Two Chinachem Exchange Square, 338 King's Road, North Point, Hong Kong, as soon as possible but in any event not less than 48 hours before the time appointed for the holding of the EGM or any adjournment thereof. Completion and return of the proxy form shall not preclude you from attending and voting in person.

This circular and the form of proxy will remain on the Stock Exchange's website at www.hkexnews.hk on the "Latest Listed Company Information" page for at least 7 days from the date of its posting. This circular will also be posted on the Company's website at http://www.legendaryedu.com.

# CONTENTS

	Page
CHARACTERISTICS OF GEM	ii
DEFINITIONS	1
LETTER FROM THE BOARD	5
APPENDIX – 2023 SHARE OPTION SCHEME	I-1
NOTICE OF EGM	EGM-1

#### **CHARACTERISTICS OF GEM**

GEM has been positioned as a market designed to accommodate small and mid-sized companies to which a higher investment risk may be attached than other companies listed on the Stock Exchange. Prospective investors should be aware of the potential risks of investing in such companies and should make the decision to invest only after due and careful consideration. Given that the companies listed on GEM are generally small and mid-sized companies, there is a risk that securities traded on GEM may be more susceptible to high market volatility than securities traded on the Main Board and no assurance is given that there will be a liquid market in the securities traded on GEM.

In this circular, unless the context otherwise requires, the following expressions shall have the following meanings:

"1% Individual Limit" has the meaning given to it in Clause 9.3 of the Appendix

to this circular

"2022 EGM" has the meaning given to it in "Letter of the Board

Adoption of the 2023 Share Option Scheme and
 Termination of the Existing Share Option Scheme" of this

circular

"2023 Share Option Scheme" the share option scheme of the Company proposed to be

approved and adopted by the Shareholders at the EGM, as amended from time to time in accordance with the

provisions thereof

"associate(s)" has the meaning given to it under the GEM Listing Rules

"Adoption Date" the date of approval and adoption of the 2023 Share Option

Scheme by the Shareholders

"Articles of Association" the articles of association of the Company

"Auditor" the auditors for the time being and from time to time of the

Company

"Board" the board of Directors

"Business Day" a day on which the Stock Exchange is open for the business

of dealing in securities

"chief executive" has the meaning given to it under the GEM Listing Rules

"Clawback" in respect of any Shares under the Option allocated or

awarded to a Selected Participant, the return or repayment of all or a specified part of such Shares by such Selected Participant and/or the ceasing or variation of the Selected Participant's entitlement to receive or be vested with all or a specified part of any such Shares under the options which have not yet been vested in the Selected Participant,

pursuant to Clause 5.13 of the Appendix to this circular

"close associate(s)" has the meaning given to it under the GEM Listing Rules

"Company" Legendary Education Group Limited, a company

incorporated in the Cayman Islands with limited liability and the Shares of which are listed on the GEM (Stock

Code: 8195)

"connected person(s)" has the meaning given to it under the GEM Listing Rules

"Consultation Conclusions" the Consultation Conclusions on Proposed Amendments to

the GEM Listing Rules relating to Share Schemes of Listed Issuers and Housekeeping Rule Amendment published by

the Stock Exchange in July 2022

"core connected person(s)" has the meaning given to it under the GEM Listing Rules

"Director(s)" the director(s) of the Company

"Eligible Person(s)" an individual or entity who may be eligible to participate

in the 2023 Share Option Scheme, which includes directors (including executive directors, non-executive directors and independent non-executive directors), officers and employees (whether full time or part time) of any member of the Group (including nominees and/or trustees of any employee benefit trust established or them). The eligibility of the Eligible Person(s) will be determined by the Board or its delegate based on their potential and/or actual

contribution to the business and development of the Group

"EGM" the extraordinary general meeting of the Company to be

convened to consider and, if thought fit, to approve the adoption of the 2023 Share Option Scheme and termination

of the Existing Share Option Scheme

"Existing Share Option Scheme" the share option scheme of the Company adopted on 25

September 2014

"GEM" GEM of the Stock Exchange

"GEM Listing Rules" the Rules Governing the Listing of Securities on GEM

"Grantee(s)" any Selected Participant who accepts the Offer in

accordance with the terms of the 2023 Share Option

Scheme

"Group" the Company and its subsidiaries

"HK\$" Hong Kong dollars, the lawful currency of Hong Kong "Hong Kong" the Hong Kong Special Administrative Region of PRC "Latest Practicable Date" 14 August 2023, being the latest practicable date prior to the printing of this circular for the purpose of ascertaining certain information contained herein "Notice" the notice convening the EGM contained in this circular "Offer" an offer for the grant of an Option made in accordance with the 2023 Share Option Scheme "Offer Date" the date on which an Offer is made to a Selected Participant, which must be a Business Day "Option(s)" option(s) to subscribe for Shares pursuant to the 2023 Share Option Scheme "Option Period" in respect of any particular Option, a period (which shall not exceed ten (10) years from the Offer Date) to be determined and notified by the Board to the Grantee thereof and, in the absence of such determination, from the date of acceptance of the Offer of such Option to the earlier of the date on which such Option lapses under the provisions of the 2023 Share Option Scheme and ten (10) years from the Offer Date "Personal Representative(s)" the person or persons who, in accordance with the laws of succession applicable in respect of the death of a Grantee, is or are entitled to exercise the Option granted to such Grantee (to the extent not already exercised) "Remuneration Committee" the remuneration committee of the Company "Scheme Mandate Limit" has the meaning given to it in Clause 9.1(a) of the Appendix to this circular "Scheme Period" a period of ten (10) years commencing on the Adoption Date "Selected Participant(s)" any Eligible Person(s) approved for participation in the Scheme and who has been granted any Options in accordance with the Scheme

"Share(s)" share(s) of HK\$0.0005 each (or of such other nominal amount as comprising the ordinary share capital of

the Company as shall result from a sub-division or a consolidation of such shares from time to time) in the

capital of the Company

"Shareholder(s)" holder(s) of the Share(s) of the Company

"Share Award Scheme" the share award scheme of the Company adopted on 25

November 2022

"Share Scheme(s)" share option scheme(s) and/or share award scheme(s)

involving issuance of new Shares adopted and to be adopted

by the Company from time to time

"Stock Exchange" The Stock Exchange of Hong Kong Limited

"Subscription Price" the price per Share at which a Grantee may subscribe for

Shares on the exercise of an Option pursuant to the 2023

Share Option Scheme

"Vesting Date" in relation to any Grantee, the earliest date on which the

Option (or a tranche thereof) granted to him/her may be exercised by such Grantee, pursuant to which Shares (or separate tranches of Shares) may be subscribed for pursuant

to the terms of such Options

"Vesting Period" in relation to any Grantee, the period commencing on the

date on which the Grantee accepts the Option granted to him/her and ending on the Vesting Date (both dates

inclusive)

"%" per cent



#### LEGENDARY EDUCATION GROUP LIMITED

# 傳承教育集團有限公司

(formerly known as Legendary Group Limited 創天傳承集團有限公司) (incorporated in the Cayman Islands with limited liability)

(Stock Code: 8195)

Executive Directors:

Mr. Yuen Yu Sum (Chairman)

Mr. Chan Lap Jin Kevin

Non-executive Directors:

Mr. Law Wing Chung

Dr. Tang Sing Hing Kenny

Independent Non-executive Directors:

Mr. Chung Chin Kwan

Mr. Chan Kim Fai Eddie

Mr. Chung Kwok Pan

Registered Office:

Maples Corporate Services Limited

P.O. Box 309, Ugland House

Grand Cayman, KY1-1104

Cayman Islands

Head office and principal place of

business in Hong Kong:

5/F, World Interests Building

8 Tsun Yip Lane

Kwun Tong

Kowloon

Hong Kong

16 August 2023

To the Shareholders

Dear Sir or Madam.

# ADOPTION OF THE 2023 SHARE OPTION SCHEME AND TERMINATION OF THE EXISTING SHARE OPTION SCHEME; AND NOTICE OF EXTRAORDINARY GENERAL MEETING

#### INTRODUCTION

The purpose of this circular is to provide you with (i) information in relation to the resolutions to be proposed at the EGM for the adoption of the 2023 Share Option Scheme and the termination of the Existing Share Option Scheme; and (ii) the notice of the EGM.

# ADOPTION OF THE 2023 SHARE OPTION SCHEME AND TERMINATION OF THE EXISTING SHARE OPTION SCHEME

The Existing Share Option Scheme was adopted by the Company on 25 September 2014 and is valid and effective for a period of 10 years from the date of adoption. Pursuant to the Consultation Conclusions, Chapter 23 of the GEM Listing Rules have been amended with effect from 1 January 2023. In light of the above, the Company proposes to terminate the Existing Share Option Scheme and adopt the 2023 Share Option Scheme to replace the Existing Share Option Scheme.

As at the Latest Practicable Date, the Company had granted Options for the subscription of a total of 465,786,640 Shares (as adjusted) under the Existing Share Option Scheme, of which 354,600,000 Options had been exercised; 44,000,000 Options granted were cancelled; and 67,186,640 Options remained outstanding.

Details of the outstanding Options as at the Latest Practicable Date are set out below:

Name of grantees	Position in the Group	Exercise price	Exercise period	Vesting conditions	Number of outstanding Options
Directors					
Mr. Yuen Yu Sum	Executive Director and Chairman	HK\$1.004	14 October 2022 to 13 October 2032	N/A	3,942,400
Mr. Chan Lap Jin Kevin	Executive Director	HK\$1.004	14 October 2022 to 13 October 2032	N/A	3,942,400
Mr. Law Wing Chung	Non-executive Director	HK\$1.004	14 October 2022 to 13 October 2032	N/A	800,800
					8,685,600
Substantial Shareholder					
Mr. Lui Yu Kin	Substantial Shareholder and tutor of an operating	HK\$1.500	26 March 2021 to 25 March 2031	N/A	3,373,920
	subsidiary of the Group, providing financial quotient and investment experience-sharing training courses	HK\$1.004	14 October 2022 to 13 October 2032	N/A	3,942,400
					7,316,320

Name of grantees	Position in the Group	Exercise price	Exercise period	Vesting conditions	Number of outstanding Options
Other 25 employees					
	Employees of the Company or its subsidiaries (other than Directors and chief executives)	HK\$1.500	26 March 2021 to 25 March 2031	N/A	24,784,320
		HK\$1.004	14 October 2022 to 13 October 2032	N/A	14,612,800
		HK\$1.004	14 October 2022 to 13 October 2032	A total of 8,258,000 share options granted to three employees will only be vested if each of the net profit of the Group for the financial year of the Company ending 31 March 2023 and 31 March 2024 is equal to or exceeds the net profit of the Group for the financial year ended 31 March 2022	9,083,800

48,480,920

Name of grantees	Position in the Group	Exercise price	Exercise period	Vesting conditions	Number of outstanding Options
Consultants					
Mr. Chan Chun Kit Jack Eason	External consultant providing marketing-related services for YH Group Limited, a principal subsidiary of the Company	HK\$1.004	14 October 2022 to 13 October 2032	A total of 2,458,000 Share Options granted to four consultants will only be vested if each of the net profit of the Group for the financial year	400,400
Mr. Tam Jeffrey Chun Yip	External consultant providing marketing-related services for Maxi Chain Enterprises Limited, a principal subsidiary of the Company	HK\$1.004	14 October 2022 to 13 October 2032	of the Company ending 31 March 2023 and 31 March 2024 is equal to or exceeds the net profit of the Group for the financial year ended 31 March 2022	701,800
Mr. Wong Siu Ki	External consultants providing mergers and acquisitions services to the	HK\$1.004	14 October 2022 to 13 October 2032		501,600
Mr. Chow Kut Fu	Company, including but not limited to identifying appropriate acquisition targets and facilitating negotiation process with sellers	HK\$1.004	14 October 2022 to 13 October 2032		1,100,000
					2,703,800
			Total		67,186,640

The Directors confirm that prior to the EGM, they will not grant any further Option under the Existing Share Option Scheme. There is no other share scheme of the Company besides the Existing Share Option Scheme and the Share Award Scheme.

According to the terms of the Existing Share Option Scheme, the Company may by an ordinary resolution in general meeting terminate the operation of the Existing Share Option Scheme, and in such event, no further options can be granted under the Existing Share Option Scheme.

Upon the termination of the Existing Share Option Scheme, no further share options would be offered under the Existing Share Option Scheme but the Existing Share Option Scheme would in other respects remain in force to the extent necessary to give effect to the exercise of the outstanding Existing Options. Any outstanding Existing Options will continue to be valid and exercisable in accordance with the terms of the Existing Share Option Scheme.

Ordinary resolutions will be proposed at the EGM for the Shareholders to consider, and if thought fit, to approve the termination of the Existing Share Option Scheme and adoption of the 2023 Share Option Scheme. The 2023 Share Option Scheme complies with the latest requirements under Chapter 23 of the GEM Listing Rules.

The Scheme Mandate Limit was approved by the Shareholders in the extraordinary general meeting held on 25 November 2022 (the "2022 EGM") with the Share Award Scheme. The maximum number of the Shares which may be issued in respect of all Options and awards to be granted under all Share Schemes of the Company (including under the 2023 Share Option Scheme, the Share Award Scheme and any other Share Scheme(s)) shall not in aggregate exceed 10% of the Shares in issue as at the 2022 EGM (i.e. 35,845,605 Shares).

None of the Directors is a trustee of the Existing Share Option Scheme and 2023 Share Option Scheme or has any direct or indirect interest in the trustees of the Existing Share Option Scheme and 2023 Share Option Scheme, if any. Other than the Existing Share Option Scheme and the Share Award Scheme, the Company did not maintain any other Share Scheme as at the Latest Practicable Date.

As at the Latest Practicable Date, to the best knowledge of the Directors and having made all reasonable enquiries, no Shareholder has any material interest in the proposed adoption of the 2023 Share Option Scheme. As such, no Shareholder is required to abstain from voting on the resolutions in relation thereto.

As at the Latest Practicable Date, none of the Directors, substantial Shareholders and their respective close associates competes or may compete with the business of the Group and any other conflict of interest which any such person has or may have with the Group.

As at the Latest Practicable Date, the Board has no present intention to grant Options to any Eligible Participants in the coming 12 months under the 2023 Share Option Scheme.

#### EXPLANATION OF THE TERMS OF THE 2023 SHARE OPTION SCHEME

Please refer to the Appendix to this circular for the terms of 2023 Share Option Scheme.

#### **Purpose**

The purpose of the 2023 Share Option Scheme is to enable the Company to grant Options to Eligible Persons as incentives or rewards for their contribution or potential contribution to the Group and/or to recruit and retain high calibre Eligible Persons and attract human resources that are valuable to the Group.

#### Duration

Subject to the provisions thereof, the 2023 Share Option Scheme shall be valid and effective until the date of expiry of the Scheme Period, after which period no further Options shall be offered or granted but the provisions of the 2023 Share Option Scheme shall remain in full force and effect to the extent necessary to give effect to any Options granted or exercised prior thereto or otherwise as may be required in accordance with the 2023 Share Option Scheme. Options complying with the provisions of the GEM Listing Rules which are granted during the life of the 2023 Share Option Scheme shall continue to be valid and exercisable in accordance with their terms of grant and provisions of the 2023 Share Option Scheme after the expiry of the Scheme Period.

#### Conditions precedent of the 2023 Share Option Scheme

The 2023 Share Option Scheme is conditional upon:

- (a) the GEM Listing Committee granting approval for the listing of, and permission to deal in, the Shares on GEM, which Shares may fall to be issued pursuant to the exercise of the Options granted under the 2023 Share Option Scheme;
- (b) the passing of an ordinary resolution by the Shareholders in general meeting to approve and adopt the 2023 Share Option Scheme and to authorise the Directors to grant Options and to allot, issue and deal with from time to time such number of Shares to subscribe for Shares hereunder pursuant to the exercise of any Options granted under the 2023 Share Option Scheme; and
- (c) the passing of an ordinary resolution by the Shareholders in general meeting to terminate the Existing Share Option Scheme.

If the above condition (a) is not satisfied on or before the date following six (6) months after the Adoption Date (or such later date as the Board may determine), the 2023 Share Option Scheme shall forthwith determine, and any Option granted or agreed to be granted pursuant to the 2023 Share Option Scheme and any offer of such grant shall be of no effect and no person shall be entitled to any rights or benefits or be under any obligations under or in respect of the 2023 Share Option Scheme or any such Option.

#### **Eligible Persons**

The participants of the 2023 Share Option Scheme shall be the Eligible Persons.

The basis of eligibility of any Eligible Person(s) to the grant of any Option shall be determined by the Board (or where necessary, the independent non-executive Directors and/or the Remuneration Committee) from time to time at its absolute discretion on the basis of the Eligible Person's contribution or potential contribution to the Group, or such other factors as the Board may deem appropriate. In determining whether a person has contributed or will contribute to the Group, the Board will take into account, among other things, whether contribution has been made to or will be made to the Group in terms of operation, financial performance, prospects, growth, development, reputation and image of the Group. In assessing the eligibility of any Eligible Person, the Board will consider all relevant factors as appropriate, including, among others, (a) his/her responsibilities and contributions made or to be made to the Group; (b) his/her ability to further promote the development of the Group; (c) his/her years of service; and (d) his/her professional qualifications and knowledge in the industry.

#### **Grant of Options**

The Directors shall, in accordance with the provisions of the 2023 Share Option Scheme, be entitled but shall not be bound, at any time within a period of ten (10) years commencing on the Adoption Date to make an Offer to any Eligible Person(s) as the Directors may in their absolute discretion select, to take up an Option pursuant to which such Eligible Person may, during the Option Period, subscribe for such number of Shares (being, subject to Clauses 9 and 10 of the Appendix to this circular, a board lot for dealing in Shares on the Stock Exchange or an integral multiple thereof) at the Subscription Price as the Directors shall, subject to Clause 6 of the Appendix to this circular, determine.

An Offer shall be deemed to have been accepted and to have taken effect when the duplicate letter comprising acceptance of the Offer duly signed by the Selected Participant, together with a payment of HK\$1.00 by way of consideration for the grant thereof, is received by the Company within twenty-eight (28) days from the Offer Date. If such payment is not enclosed, the acceptance shall not be prejudiced thereby, but the acceptance shall create a promise by the Selected Participant to pay to the Company HK\$1.00 on demand. Such payment shall in no circumstances be refundable.

#### **Vesting Period**

The Vesting Period for any Option granted to any Grantee under the 2023 Share Option Scheme shall not be less than twelve (12) months from the date of grant of such Option. The Board or the Remuneration Committee shall have the authority to determine a shorter Vesting Period on Options in its sole discretion under each of the following circumstances:

- (a) grants of "make-whole" Options to a Selected Participant who newly joined the Group to replace the share options he/she forfeited when leaving the previous employers;
- (b) grants to a Selected Participant whose employment is terminated due to disability or occurrence of any out of control event;
- (c) grants that are made in batches during a year for administrative and compliance reasons, which may include Options that should have been granted earlier but had to wait for a subsequent batch, in such cases, the Vesting Date may be adjusted to take account of the time from which the Options would have been granted if not for such administrative or compliance requirements;
- (d) grants of Options with a mixed or accelerated vesting schedule such that the Options may vest evenly over a period of twelve (12) months; and
- (e) grants of Options with a total vesting and holding period of more than twelve (12) months.

Such discretion gives the Company more flexibility to (i) adapt to exceptional and justified circumstances; or (ii) attract talents or reward exceptional performers with accelerated vesting. These circumstances are also considered by the Stock Exchange to be justifiable reasons for having a shorter vesting period as set out in the Consultation Conclusions.

Accordingly, the Directors (including the independent non-executive Directors) are of the view that the discretion in allowing a shorter vesting period in each of the circumstances as detailed above is appropriate and in line with the purpose of the 2023 Share Option Scheme (and other share scheme(s), where applicable). There is no performance target which must be achieved by the Grantee before an Option can be exercised under the terms of the 2023 Share Option Scheme save as otherwise imposed by the Board as it thinks fit in the relevant Offer.

#### **Option Period**

The Option Period of an Option shall not exceed ten (10) years from the Offer Date.

#### Performance targets

The rules of the 2023 Share Option Scheme will not prescribe specific performance targets that must be met before an Option can be exercised. However, the rules of the 2023 Share Option Scheme will give the Directors absolute discretion to impose such conditions on the Share Options where appropriate. The Directors consider that, notwithstanding that it may not always be appropriate to impose such conditions particularly when the purpose of granting the Options is to remunerate or compensate Selected Participants for past contributions, it would be more beneficial to the Company to retain the flexibility to determine whether such conditions are appropriate in light of the particular circumstances of each grant. If performance targets are imposed on a Grantee upon the grant of Options, the Board shall have regard to the purpose of the 2023 Share Option Scheme in assessing such performance targets, with reference to factors including but not limited to, as and when appropriate, sales performance (e.g. revenue), operating performance (e.g. profits, operation efficiency in terms of cost control), financial performance (e.g. profits, cash flow, earnings, market capitalisation, return on equity) of the Group, corporate sustainability parameter (e.g. timeliness and accuracy in handling customer feedback, team work capabilities, adherence to corporate culture) and discipline and responsibility (e.g. punctuality, integrity, honesty or compliance with internal business procedures), the satisfaction of which shall be assessed and determined by the Board at its sole discretion. The Company shall utilise its internal assessment system to appraise and evaluate the performance targets applicable to each grant of Options on a case-by-case basis. The Company shall consider the past contributions of a Selected Participant with reference to the factors set out above and form an internal assessment as regards the future value that such Selected Participant may bring to the growth and development of the Group. The assessment shall involve the consideration and appraisal of the Selected Participant's expected contribution with reference to such Selected Participant's nature of duties (e.g. whether in a sale role, management role or a support role), position within the Group (e.g. whether overall Group level targets or specific performance indicators should be adopted) and other features including geographical location, corporate culture and business strategy focus. Specific weightings shall be given to the factors above in order to provide a fair and objective appraisal of the Selected Participants before the grant of Options, such that the grants will be on a fair and reasonable basis and in the interest of the Company and its Shareholders as a whole. The management shall propose the performance targets of each Selected Participant, if any, to the Board or the remuneration committee of the Company for consideration, who shall then assess the reasonableness and suitability of such performance targets.

#### Clawback mechanism

In certain circumstances, it may be regarded as inequitable for any Shares under the Options to be vested or retained, as applicable. Such Options are therefore subject to Clawback, including but not limited to circumstances where there has been a material misstatement or omission in the financial statements of the Group made by a Selected Participant or if the relevant Selected Participant has engaged in serious negligence, fraud or misconduct, breached the Company's policies, rules or regulations or other circumstances. Notwithstanding anything else in the 2023 Share Option Scheme, any Shares under the Options may be subject to Clawback pursuant to the Company's Clawback mechanism, as amended from time to time. The rules of the 2023 Share Option Scheme will not prescribe specific clawback mechanism to recover or withhold the Options to be granted. However, the rules of the New Share Option Scheme will give the Board discretion to impose such conditions on the Options or prescribe such clawback mechanism where appropriate. The Directors consider that, notwithstanding that it may not always be appropriate to prescribe such clawback mechanism particularly when the purpose of granting the Options is to remunerate or compensate Selected Participants for past contributions, it would be more beneficial to the Company to retain the flexibility to determine whether such conditions are appropriate in light of the particular circumstances of each grant.

If a clawback mechanism is imposed, the Board will take into account individual circumstances when devising such mechanism such as the role of the holder of the Options, the purpose of the grant (for example whether as recognition of past contribution or as incentive to motivate such holder of the options to contribute to the Group in the future on an ongoing basis), whether it would be particularly burdensome and complicated to implement the clawback mechanism, whether there are any tax implications etc. By allowing the Company to impose clawback mechanism as may be stipulated in the offer letter on a case by case basis, the Company may be in a better position to retain such Eligible Persons to continue serving the Company whilst at the same time providing these Eligible Persons further incentive in achieving the goals of the Group, and therefore aligns with the purpose of the 2023 Share Option Scheme.

#### **Determination basis of the Subscription Price**

The Subscription Price in respect of any Option shall, subject to any adjustments made pursuant to Clause 11 of the Appendix to this circular, be such price as the Directors at their absolute discretion shall determine, provided that it must be no less than the higher of:

- (a) the closing price of Shares as stated in the Stock Exchange's daily quotations sheet on the Offer Date (which must be a Business Day);
- (b) the average closing price of Shares as stated in the Stock Exchange's daily quotations sheet for the five (5) Business Days immediately preceding the Offer Date; and
- (c) the par value of the Shares on the Offer Date.

The Directors consider that such basis will serve to preserve the value of the Company and encourage the Selected Participants to acquire proprietary interests in the Company.

#### Early termination of Option Period

The Option Period in respect of any Option shall automatically terminate and that Option (to the extent not already exercised) shall lapse on the earliest of:

- (a) the expiry of the Option Period;
- (b) the expiry of any of the following periods:
  - (i) in the event of the Grantee ceases to be an Eligible Person thereof by reason of his/her death, permanent disability or retirement in accordance with his/her contract of employment and the termination of his/her employment is not on one or more of the grounds specified in (c) below before exercising the Option in full, the Grantee or, as appropriate, his/her Personal Representative(s) may exercise the Option (to the extent not already exercised) in whole or in part in accordance with the provisions of Clause 7.2 of the Appendix to this circular within a period of twelve (12) months following the date of cessation of employment which date shall be the last day on which the Grantee was at work with the member of the Group whether salary is paid in lieu of notice or not, or such longer period as the Directors may determine, and any such Options not so exercised shall lapse and determine at the end of such period;
  - (ii) in the event that the Grantee ceases to be an Eligible Person thereof by reason other than his/her death, permanent disability or retirement in accordance with his/her contract of employment or the termination of his/her employment is on one or more of the grounds specified in (c) below before exercising the Option in full, the Option (to the extent not already exercised) shall lapse on the date of cessation or termination and not be exercisable unless the Directors otherwise determine, in which event the Option (to the extent not already exercised) shall be exercisable in whole or in part within such period as the Directors may in its absolute discretion determine following the date of such cessation or termination. The date of cessation or termination as aforesaid shall be the last day on which the Grantee was actually at work with the member of the Group whether salary is paid in lieu of notice or not;

- (iii) if a general offer, whether by way of take-over offer, share re-purchase offer, or scheme of arrangement or otherwise in like manner is made to all Shareholders, and such offer becomes or is declared unconditional (in the case of take-over offer) or is approved by the requisite majorities at the relevant meeting of the Shareholders (in the case of a scheme of arrangement), the Directors shall as soon as practicable notify the Grantee (or his/her Personal Representative(s)) accordingly. The Grantee (or his/her Personal Representative(s)) shall be entitled to exercise his/her Option (to the extent not already exercised) either to its full extent or to the extent specified in the Grantee's notice to the Company at any time (in the case of takeover offer) within one (1) month after the date on which the offer becomes or is declared unconditional or (in the case of a scheme of arrangement) prior to such time and date as shall be notified by the Directors. To the extent that any Option has not been so exercised, it shall upon the expiry of such period lapse and determine;
- (iv) in the event of a notice is given by the Company to its Shareholders to convene a general meeting for the purposes of considering, and if thought fit, approving a resolution to voluntarily wind-up the Company, the Company shall on the same date as or soon after it despatches such notice to each Shareholder of the Company give notice thereof to all Grantees and thereupon, each Grantee (or his/her Personal Representative(s)) shall be entitled to exercise his/her Option (to the extent not already exercised) either to its full extent or to the extent specified in such notice in accordance with the provisions of Clause 7.2 of the Appendix to this circular at any time no later than two (2) Business Days prior to the proposed general meeting of the Company by giving notice to the Company in accordance with the provisions of Clause 7.2 of the Appendix to this circular, accompanied by a remittance of the Subscription Price for the Shares in respect of which the notice is given. The Company shall as soon as possible and, in any event, no later than the Business Day immediately prior to the date of the proposed general meeting referred to above, allot the relevant Shares to the Grantee credited as fully paid; and

- if a compromise or arrangement between the Company and its members or creditors is (v) proposed for the purposes of or in connection with a scheme for the reconstruction of the Company or its amalgamation with any other company or companies, the Company shall give notice thereof to all Grantees (together with a notice of the existence) on the same date as it despatches to each member or creditor of the Company a notice summoning the meeting to consider such a compromise or arrangement, and thereupon each Grantee (or his/her Personal Representative(s)) shall be entitled, at any time not later than (i) the Option Period, (ii) the period of two (2) months from the date of such notice, or (iii) the date on which such compromise or arrangement is sanctioned by the court, whichever is the earlier, to exercise his/her Options in whole or in part. The Company shall as soon as possible and in any event no later than one (1) Business Day prior to the date of such general meeting, allot and issue such number of Shares to the Grantees which fall to be issued on such exercise credited as fully paid. Subject thereto, all Options then outstanding shall lapse and determine upon such compromise or arrangement becoming effective;
- (c) the date on which the Grantee ceases to be an employee by reason of voluntary resignation or dismissal or upon expiration of his/her term of directorship (unless immediately renewed upon expiration), or by a termination of his/her employment on the grounds that he/she has been guilty of persistent or serious misconduct, or has committed any act of bankruptcy or has become insolvent or has made any arrangement or composition with his/her creditors generally, or has been convicted of any criminal offence (other than an offence which in the opinion of the Directors does not bring the Grantee or the Group into disrepute or otherwise does not compromise the integrity or involve dishonesty of the Grantee) or any other grounds on which the relevant member of the Group would be entitled to terminate the Grantee's employment or directorship pursuant to any applicable laws. Transfer of employment or engagement or relationship from one member of the Group to another member of the Group shall not be considered as a cessation of employment, engagement or relationship;
- (d) subject to Clause 7.3(d) of the Appendix to this circular; the date of the commencement of the winding-up of the Company; and
- (e) the date on which the Directors shall exercise the Company's right to cancel the Option by reason of breach of Clause 7.1 of the Appendix to this circular by the Grantee in respect of that or any other Option.

#### Scheme Mandate Limit and 1% Individual Limit

Subject to Clause 9.2 and Clause 11 of the Appendix to this circular, the maximum number of the Shares which may be issued in respect of all Options and awards to be granted under all Share Schemes of the Company (including under the 2023 Share Option Scheme and any other Share Scheme(s)) shall not in aggregate exceed 10% of the Shares in issue as at the Adoption Date.

The Company may seek approval by the Shareholders in general meeting for refreshing the Scheme Mandate Limit under the 2023 Share Option Scheme after three (3) years from the Adoption Date or the date of the Shareholders' approval for the last refreshment (as the case may be).

The Scheme Mandate Limit was approved by the Shareholders in the extraordinary general meeting held on 2022 EGM with the Share Award Scheme. The maximum number of the Shares which may be issued in respect of all Options and awards to be granted under all Share Schemes of the Company (including under the 2023 Share Option Scheme, the Share Award Scheme and any other Share Scheme(s)) shall not in aggregate exceed 10% of the Shares in issue as at the 2022 EGM (i.e. 35,845,605 Shares).

Any refreshment within any three-year period must be approved by Shareholders (other than the controlling Shareholders (or if there is no controlling Shareholder, the Directors (excluding independent non-executive Directors) and chief executive of the Company and their respective associates) and their associates, who must abstain from voting in favour of the relevant resolution at the general meeting). The total number of Shares which may be issued in respect of all options and awards to be granted under all Share Scheme(s) (including the 2023 Share Option Scheme) under the Scheme Mandate Limit as refreshed shall not exceed 10% of the relevant class of Shares in issue as at the date of approval of the refreshed Scheme Mandate Limit by Shareholders in general meeting.

The Company may seek separate approval by the Shareholders in general meeting for granting Options beyond the Scheme Mandate Limit provided the Options in excess of the Scheme Mandate Limit are granted only to Eligible Persons specifically identified by the Company before such approval is sought. The Company must sent a circular to Shareholders containing the name of each specified Eligible Persons who may be granted such Options, the number and terms of the Options to be granted to each Eligible Person, the purpose of granting Options to the specified Eligible Persons with an explanation as to how the terms of the Options serve such purpose. The number and terms of Options to be granted to such Eligible Persons must be fixed before Shareholders' approval.

The total number of Shares issued and to be issued in respect of all Options and awards granted to each Selected Participant under the 2023 Share Option Scheme and any other Share Scheme(s) (excluding options and awards that have been lapsed in accordance with the terms of the 2023 Share Option Scheme and any other Share Scheme(s)) in any 12-month period up to and including the date of such grant shall not in aggregate exceed 1% Individual Limit. Where any grant of Options to an Eligible Person would result in the Shares issued and to be issued in respect of all Options and awards granted and to be granted to such Eligible Person (excluding options and awards that have been lapsed in accordance with the 2023 Share Option Scheme and any other Share Scheme(s)) in excess of the 1% Individual Limit in any 12-month period up to and including the date of such grant, such grant must be separately approved by the Shareholders in general meeting with such Eligible Person and his/her close associates (or his/her associates if the Eligible Person is a connected person) abstaining from voting.

#### **Grant of Options to connected persons**

Any grant of Options to a director, chief executive or substantial shareholder of the Company or any of their respective associates must be approved by the independent non-executive Directors (excluding any independent non-executive Director who is a Grantee) and shall comply with the relevant provisions of Chapter 23 of the GEM Listing Rules.

Where any grant of Options to a substantial shareholder or an independent non-executive Director or any of their respective associates would result in the Shares issued and to be issued in respect of all Options under the 2023 Share Option Scheme and all options, awards and securities granted under other Share Scheme(s) (including Options exercised, cancelled and outstanding but excluding those Options, awards and securities lapsed in accordance with the terms of such schemes) to that person in the 12-month period up to and including the date of such grant representing in aggregate over 0.1% of the total number of Shares in issue, such grant of Options must be approved by the Shareholders in general meetings in the manner set out in Rule 23.04(4) of the GEM Listing Rules. The Grantee, his/her associates and all core connected persons of the Company must abstain from voting in favour at such general meeting.

#### Voting rights, dividend, transfer or any other rights

The Options do not carry any right to vote in general meeting of the Company, or any right, dividend, transfer or any other rights, including those arising on the liquidation of the Company.

#### Ranking of the Shares to be allotted and issued upon the exercise of an Option

The Shares to be allotted and issued upon the exercise of an Option will be subject to all the provisions of the memorandum of association of the Company and the Articles for the time being in force and will rank pari passu in all respects with the then existing fully paid Shares in issue on the Allotment Date and accordingly will entitle the holders thereof to participate in all dividends or other distributions paid or made on or after the Allotment Date other than any dividend or other distribution previously declared or recommended or resolved to be paid or made if the record date therefore shall be before the Allotment Date. A Share allotted and issued upon the exercise of an Option shall not carry voting rights until the name of the Grantee has been duly entered on the register of members of the Company as the holder thereof.

#### **Transferability**

An Option shall be personal to the Grantee and shall not be transferable or assignable and no Grantee shall in any way sell, transfer, charge, mortgage, encumber or otherwise dispose of or create any interest (whether legal or beneficial) whatsoever in favour of any third party over or in relation to any Option or enter into any agreement to do so.

#### Adjustment arising from reorganisation of capital structure

In the event of any alteration in the capital structure of the Company whilst any Option remains exercisable or this Scheme remains in effect, and such event arises from a capitalisation of profits or reserves, rights issue, consolidation, sub-division or reduction of the share capital of the Company, then, in any such case, the Company shall instruct the Auditors or the independent financial adviser to the Company to certify in writing the adjustment, if any, to be made either generally or as regards any particular Grantee, to:

- (a) the number of Shares to which this Scheme or any Option(s) relates (insofar as it is/they are unexercised); and/or
- (b) the Subscription Price of any unexercised Option; and/or
- (c) the number of Shares in respect of which any further Options may be granted within the 1% Individual Limit and/or the Scheme Mandate Limit,

and an adjustment as so certified by the Auditors or the independent financial adviser to the Company shall be made, provided that:

- (i) the aggregate Subscription Price payable by a Grantee on the full exercise of any Option shall remain as nearly as possible the same (but shall not be greater than) as it was before such event;
- (ii) no such adjustment shall be made the effect of which would be to enable a Share to be issued at less than its nominal value;
- (iii) no such adjustment shall be made the effect of which would increase the proportion of the issued share capital of the Company, rounded to the nearest whole Share, for which any Grantee would have been entitled to subscribe had he/she exercised all the Options held by him/her immediately prior to such adjustment;
- (iv) the issue of Shares or securities of the Company as consideration in a transaction shall not be regarded as a circumstance requiring any such adjustment; and
- (v) for the avoidance of doubt, any adjustment shall be made in compliance with the GEM Listing Rules or other relevant guidance as the Stock Exchange may from time to time issue.

In respect of any such adjustments, other than any made on a capitalisation issue, the Auditors or the independent financial adviser to the Company must confirm to the Directors in writing that such adjustments satisfy the requirements set out in rule 23.03(13) of the GEM Listing Rules.

#### Alteration of the 2023 Share Option Scheme

Any change to the terms of Options granted to a Selected Participant must be approved by the Board, the Remuneration Committee, the independent non-executive Directors and/or the Shareholders (as the case may be) if the initial grant of the Options was approved by the Board, the Remuneration Committee, the independent non-executive Directors and/or the Shareholders (as the case may be), except where the alterations take effect automatically under the existing terms of the 2023 Share Option Scheme.

Any alterations to the terms and conditions of the 2023 Share Option Scheme which are of a material nature or any alterations to the provisions relating to the matters set out in Rule 23.03 of the GEM Listing Rules to the advantage of Selected Participants must be approved by the Shareholders in general meeting.

Any change to the authority of the Directors or the scheme administrators in relation to any alteration of the terms of the 2023 Share Option Scheme must be approved by the Shareholders in general meeting.

The amended terms of the 2023 Share Option Scheme or the Options must still comply with the relevant requirements under the GEM Listing Rules.

#### **Termination**

The Company may by resolution in general meeting at any time terminate the operation of the 2023 Share Option Scheme and in such event no further Options will be offered or granted but in all other respects the provisions of the 2023 Share Option Scheme shall remain in force to the extent necessary to give effect to the exercise of any outstanding Options granted prior thereto or otherwise as may be required in accordance with the provisions of the 2023 Share Option Scheme and outstanding Options granted prior to such termination shall continue to be valid and exercisable in accordance with the provisions of the 2023 Share Option Scheme.

#### **Cancellation of Options granted**

The Directors may effect the cancellation of any Options granted but not exercised on such terms as may be agreed with the relevant Grantee, as the Directors may in their absolute discretion see fit and in a manner that complies with all applicable legal requirements for cancellation. Where the Company cancels any Options granted and make a new grant of Options to the same Grantee, such new grant of Options may only be made under the 2023 Share Option Scheme if there are available unissued Options (excluding the cancelled Options) under the 2023 Share Option Scheme within the Scheme Mandate Limit.

Options cancelled in accordance with the terms of the 2023 Share Option Scheme will be regarded as utilised for the purpose of calculating the Scheme Mandate Limit. Options lapsed in accordance with the terms of the 2023 Share Option Scheme will not be regarded as utilised for the purpose of calculating the Scheme Mandate Limit.

#### REASONS FOR THE ADOPTION OF THE 2023 SHARE OPTION SCHEME

The 2023 Share Option Scheme, as a replacement of the Existing Share Option Scheme, will allow the Company to continue to provide equity incentives to Eligible Persons so as to continue to serve its intended purposes of aligning the interests of Eligible Persons with those of the Group through ownership of Shares, dividends and other distributions paid on Shares and/or the increase in value of the Shares and to encourage and enable Eligible Persons to make contributions to the long-term growth and profits of the Group and share the success of the Group.

The Directors therefore consider that the adoption of the 2023 Share Option Scheme is in the interests of the Company and the Shareholders as a whole.

#### **EGM**

The EGM will be convened and held on Friday, 1 September 2023 at 11:00 a.m. at 5/F, World Interests Building, 8 Tsun Yip Lane, Kwun Tong, Kowloon, Hong Kong to consider and, if thought fit, approve the adoption of the 2023 Share Option Scheme and termination of the Existing Share Option Scheme. In compliance with the GEM Listing Rules, the resolutions will be voted on by way of poll at the EGM. The poll results will be announced in accordance with Rule 17.47(5) of the GEM Listing Rules after the conclusion of the EGM.

A notice convening the EGM is set out on pages EGM-1 to EGM-3 of this circular. A form of proxy for use by the Shareholders at the EGM is enclosed. Whether or not you intend to attend the EGM in person, you are requested to complete the enclosed form of proxy in accordance with the instructions printed thereon and return it to the branch share registrar and transfer office of the Company in Hong Kong, Union Registrars Limited at Suites 3301-04, 33/F., Two Chinachem Exchange Square, 338 King's Road, North Point, Hong Kong, as soon as possible and in any event no later than 48 hours before the time appointed for holding the EGM or any adjournment thereof. Completion and return of the form of proxy will not preclude you from attending and voting in person at the EGM or any adjournment of it, if you so wish.

#### CLOSURE OF REGISTER OF MEMBERS

For determining the entitlement to attend and vote at the EGM, the register of members of the Company will be closed from Tuesday, 29 August 2023 to Friday, 1 September 2023, both dates inclusive, during which period no transfer of Shares will be registered. In order to be eligible to attend and vote at the EGM, unregistered holders of Shares shall ensure that all transfer documents accompanied by the relevant Share certificates must be lodged for registration with the branch share registrar and transfer office of the Company in Hong Kong, Union Registrars Limited at Suites 3301-04, 33/F., Two Chinachem Exchange Square, 338 King's Road, North Point, Hong Kong, no later than 4:00 p.m. on Monday, 28 August 2023.

#### **DOCUMENT ON DISPLAY**

A copy of the 2023 Share Option Scheme will be published on the respective websites of the Stock Exchange (www.hkexnews.hk) and the Company (www.legendaryedu.com) for display for a period of not less than 14 days before the date of EGM and will be made available for inspection at the EGM.

#### RESPONSIBILITY STATEMENT

This circular, for which the Directors collectively and individually accept full responsibility, includes particulars given in compliance with the GEM Listing Rules for the purpose of giving information with regard to the Company. The Directors, having made all reasonable enquiries, confirm that, to the best of their knowledge and belief, the information contained in this circular is accurate and complete in all material respects and not misleading or deceptive, and there are no other matters the omission of which would make any statement herein or this circular misleading.

#### RECOMMENDATION

The Directors consider that the proposed adoption of the 2023 Share Option Scheme and termination of the Existing Share Option Scheme as mentioned above are in the best interests of the Company and the Shareholders as a whole. Accordingly, the Directors recommend that all Shareholders should vote in favour of the relevant resolutions to be proposed at the EGM.

By Order of the Board **Legendary Education Group Limited Yuen Yu Sum** 

Chairman and Executive Director



#### LEGENDARY EDUCATION GROUP LIMITED

# 傳承教育集團有限公司

(formerly known as Legendary Group Limited 創天傳承集團有限公司)
(incorporated in the Cayman Islands with limited liability)
(Stock Code: 8195)

# RULES RELATING TO THE SHARE OPTION SCHEME

# ADOPTED BY A RESOLUTION PASSED BY THE SHAREHOLDERS AT THE GENERAL MEETING HELD ON 1 SEPTEMBER 2023

The purpose of this Scheme is to grant Options to Eligible Persons as incentives or rewards for their contribution or potential contribution to the Group and/or to recruit and retain high calibre Eligible Persons and attract human resources that are valuable to the Group.

#### 1. **DEFINITIONS**

1.1 In this Scheme the following expressions, except where the context otherwise requires, shall have the following meanings:

"1% Individual Limit" has the meaning given to it in Clause 9.3;

"2014 Share Option Scheme" means the share option scheme of the Company

adopted on 25 September 2014

"Adoption Date" means 1 September 2023, the date on which this

Scheme is amended and adopted by an ordinary

resolution of the Shareholders;

"Allotment Date" has the meaning given to it in Clause 7.4;

"Articles" means the articles of association of the Company

(as amended from time to time);

"associates" has the meaning given to it under the GEM

Listing Rules;

#### 2023 SHARE OPTION SCHEME

"Auditors"	means	the	auditors	for	the	time	being	and	from

time to time of the Company;

"Board" means the board of Directors;

"Business Day" means a day on which the Stock Exchange is open

for the business of dealing in securities;

"Clawback" means, in respect of any Shares under the Option

allocated or awarded to a Selected Participant, the return or repayment of all or a specified part of such Shares by such Selected Participant and/or the ceasing or variation of the Selected Participant's entitlement to receive or be vested with all or a specified part of any such Shares under the options which have not yet been vested in the Selected Participant, pursuant to Clause

5.13;

"close associate" has the meaning given to it under the GEM

Listing Rules;

"Companies Ordinance" means the Companies Ordinance (Chapter

622 of the Laws of Hong Kong), as amended, supplemented and modified from time to time;

"Company" means Legendary Education Group Limited 傳承

教育集團有限公司, a company incorporated under the laws of the Cayman Islands on 5 June 2014, the shares of which are listed on GEM (Stock

Code: 8195);

"connected person" has the meaning given to it under the GEM

Listing Rules;

"core connected person" has the meaning given to it under the GEM

Listing Rules;

"Director(s)" means the director(s) of the Company from time

to time;

"Offer"

## **2023 SHARE OPTION SCHEME**

means an offer for the grant of an Option made in

accordance with Clause 5.3;

"Eligible Person(s)"	means an individual or entity who may be eligible to participate in this Scheme, which includes Directors (including executive Directors, non-executive Directors and independent non-executive Directors), officers and employees (whether full time or part time) of any member of the Group (including nominees and/or trustees of any employee benefit trust established or them). The eligibility of the Eligible Person(s) will be determined by the Board or its delegate based on their potential and/or actual contribution to the business and development of the Group;
"GEM"	means GEM of the Stock Exchange;
"GEM Listing Committee"	has the meaning given to it in GEM Listing Rules;
"GEM Listing Rules"	means the Rules Governing the Listing of Securities on GEM of The Stock Exchange of Hong Kong Limited (as modified from time to time);
"Grantee"	means any Selected Participant who accepts or is deemed to have accepted the Offer in accordance with the terms of this Scheme or (where the context so permits and as referred to in Clause 7.3(a)) the Personal Representative of such person;
"Group"	means the Company and its Subsidiaries from time to time;
"HK\$"	means Hong Kong dollars, the lawful currency of Hong Kong;
"Hong Kong"	means the Hong Kong Special Administrative Region of the People's Republic of China;
((a) a)	

#### 2023 SHARE OPTION SCHEME

"Offer Date" means the date on which an Offer is made to a

Selected Participant, which must be a Business

Day;

"Option" means an option to subscribe for a specified

number of Shares in issue at the Subscription

Price;

"Option Period" means in respect of any particular Option, a period

(which shall not exceed ten (10) years from the Offer Date) to be determined and notified by the Board to the Grantee thereof and, in the absence of such determination, from the date of acceptance of the Offer of such Option to the earlier of the date on which such Option lapses under the provisions of Clause 8 and ten (10) years from the

Offer Date;

"Personal Representative(s)" means the person or persons who, in accordance

with the laws of succession applicable in respect of the death of a Grantee, is or are entitled to exercise the Option granted to such Grantee (to

the extent not already exercised);

"Scheme" means this share option scheme in its present form

or as amended from time to time in accordance

with the provisions hereof;

"Scheme Mandate Limit" has the meaning given to it in Clause 9.1(a);

"Scheme Period" means a period of ten (10) years commencing on

the Adoption Date;

"Selected Participant(s)" means any Eligible Person approved for

participation in this Scheme and who has been granted any Options in accordance with this

Scheme;

# **2023 SHARE OPTION SCHEME**

"Share(s)"	means share(s) of HK\$0.0005 each (or of such other nominal amount as comprising the ordinary share capital of the Company as shall result from a sub-division or a consolidation of such shares from time to time) in the capital of the Company;
"Shareholder(s)"	means holder(s) of the Share(s);
"Share Scheme(s)"	means share option scheme(s) and/or share award scheme(s) involving issuance of new Shares adopted and to be adopted by the Company from time to time;
"Stock Exchange"	means The Stock Exchange of Hong Kong Limited;
"Subscription Price"	means the price per Share at which a Grantee may subscribe for Shares on the exercise of an Option pursuant to Clause 6;
"Subsidiary(ies)"	means a company or companies which is for the time being and from time to time a subsidiary (within the meaning of the Companies Ordinance) of the Company, whether incorporated in Hong Kong or elsewhere;
"substantial shareholder"	has the meaning given to it under the GEM Listing Rules;
"Vesting Date"	means, in relation to any Grantee, the earliest date on which the Option (or a tranche thereof) granted to him/her may be exercised by such Grantee, pursuant to which Shares (or separate tranches of Shares) may be subscribed for pursuant to the terms of such Option;
"Vesting Period"	means, in relation to any Grantee, the period commencing on the date on which the Grantee accepts the Option granted to him/her and ending on the Vesting Date (both dates inclusive); and
"%"	per cent.

- 1.2 References to statutory provisions shall be construed as references to those provisions as amended or re-enacted or as their application is modified by other provisions (whether before or after the date hereof) from time to time and shall include any provisions of which they are re-enactments (whether with or without modification).
- 1.3 References herein to Clauses are to clauses of this Scheme unless the context requires otherwise.
- 1.4 The headings are inserted for convenience only and shall not affect the construction of this Scheme.
- 1.5 Reference to a person includes any public body and any body of persons, corporate or unincorporated.
- 1.6 Unless the context requires otherwise, words importing the singular include the plural and vice versa and words importing a gender or the neuter include both genders and the neuter and vice versa.

#### 2. CONDITIONS

- 2.1 This Scheme is conditional upon:
  - (a) the GEM Listing Committee granting approval for the listing of, and permission to deal in, the Shares on which may fall to be issued pursuant to the exercise of the Options granted under this Scheme;
  - (b) the passing of an ordinary resolution by the Shareholders in general meeting of the Company to approve and adopt this Scheme and to authorise the Directors to grant Options to subscribe for Shares hereunder and to allot, issue and deal with from time to time of Shares pursuant to the exercise of any Options granted under this Scheme; and
  - (c) the passing of an ordinary resolution by the Shareholders in general meeting to terminate the 2014 Share Option Scheme.

If the above Clause 2.1(a) is not satisfied on or before the date following six (6) months after the Adoption Date (or such later date as the Board may determine), this Scheme shall forthwith determine, and any Option granted or agreed to be granted pursuant to this Scheme and any offer of such grant shall be of no effect and no person shall be entitled to any rights or benefits or be under any obligations under or in respect of this Scheme or any such Option.

- 2.2 Reference in Clause 2.1(a) to the GEM Listing Committee formally granting the approvals, listing and permission referred to therein shall include any such approvals, listing and permission which are granted subject to the fulfillment of any condition precedent or condition subsequent.
- 2.3 A certificate of a Director that the conditions set out in Clause 2.1 have been satisfied and the date on which such conditions were satisfied or that such conditions have not been satisfied as of any particular date and the exact date of the Adoption Date shall be conclusive evidence of the matters certified.
- 2.4 The Company shall publish an announcement on the outcome of the Shareholders' meeting as referred to in Clause 2.1(b) for the adoption of this Scheme in the manner as set out in rule 17.47(5) of the GEM Listing Rules.

#### 3. PURPOSE, DURATION AND ADMINISTRATION

- 3.1 The purpose of this Scheme is to enable the Company to grant Options to Eligible Persons as incentives or rewards for their contribution or potential contribution to the Group and/or to recruit and retain high calibre Eligible Persons and attract human resources that are valuable to the Group.
- 3.2 Subject to Clauses 2 and 15, this Scheme shall be valid and effective until the date of expiry of the Scheme Period, after which period no further Options shall be offered or granted but the provisions of this Scheme shall remain in full force and effect to the extent necessary to give effect to any Options granted or exercised prior thereto or otherwise as may be required in accordance with the provisions of this Scheme. Options complying with the provisions of the GEM Listing Rules which are granted during the life of this Scheme shall continue to be valid and exercisable in accordance with their terms of grant and provisions of this Scheme after the expiry of the Scheme Period.
- 3.3 This Scheme shall be subject to the administration of the Board whose decision on all matters arising in relation to this Scheme or their interpretation or effect shall (save as otherwise provided herein) be final and binding on all persons who may be affected thereby. The Board shall have the right to (i) interpret and construe the provisions of this Scheme, (ii) determine the persons who will be offered Options under this Scheme, the number of Shares and the Subscription Price, subject to Clause 6, in relation to such Options, (iii) subject to Clauses 11 and 14, make such appropriate and equitable adjustments to the terms of the Options granted under this Scheme as it deems necessary, and (iv) make such other decisions or determinations as it shall deem appropriate in the administration of this Scheme.

3.4 A Grantee shall ensure and, by accepting an Offer, shall be deemed to have represented and undertaken to the Company, that the acceptance of an Offer, the holding and exercise of his/her Option in accordance with this Scheme, the allotment and issue of Shares to him/her upon the exercise of his/her Options and the holding of such Shares are and will be valid and complies and will comply with all laws, legislations and regulations including all applicable exchange control, fiscal and other laws to which he/she is subject. The Directors may, as a condition precedent to making an Offer and/ or allotting Shares upon an exercise of an Option, require a Grantee to produce such evidence as they may reasonably require for such purpose.

#### 4. DETERMINATION OF ELIGIBILITY

- 4.1 The basis of eligibility of any Eligible Person(s) to the grant of any Option shall be determined by the Board (or where necessary, the independent non-executive Directors and/or the remuneration committee of the Company) from time to time in its absolute discretion on the basis of the Eligible Person's contribution or potential contribution to the Group, or such other factors as the Board may deem appropriate. In determining whether a person has contributed or will contribute to the Group, the Board will take into account, among other things, whether contribution has been made to or will be made to the Group in terms of operation, financial performance, prospects, growth, development, reputation and image of the Group. In assessing the eligibility of any Eligible Person, the Board will consider all relevant factors as appropriate, including, among others, (a) his/her responsibilities and contributions made or to be made to the Group; (b) his/her ability to further promote the development of the Group; (c) his/her years of service; and (d) his/her professional qualifications and knowledge in the industry.
- 4.2 In order for a person to satisfy the Directors that he/she is qualified to be (or where applicable, continues to be qualified to be) an Eligible Person, such person shall provide all such information as the Directors may request for the purpose of assessing his/her eligibility (or continuing eligibility).

#### 5. GRANT OF OPTIONS

5.1 The Directors shall, in accordance with the provisions of this Scheme, be entitled but shall not be bound, at any time within a period of ten (10) years commencing on the Adoption Date to make an Offer to any Eligible Person(s) as the Directors may in their absolute discretion select, to take up an Option pursuant to which such Eligible Person may, during the Option Period, subscribe for such number of Shares (being, subject to Clauses 9 and 10, a board lot for dealing in Shares on the Stock Exchange or an integral multiple thereof) at the Subscription Price as the Directors shall, subject to Clause 6, determine.

- 5.2 The Directors shall not make an Offer after inside information has come to their knowledge until such inside information has been published in accordance with rules 16.17 to 16.19 of the GEM Listing Rules or in such manner as required under the GEM Listing Rules. In particular, no Option shall be granted during the period commencing one (1) month immediately preceding the earlier of: (i) the date of the meeting of the Directors (as the date is first notified to the Stock Exchange) for the approval of the Company's results for any year, half-year, quarterly or any other interim period (whether or not required under the GEM Listing Rules); and (ii) the deadline for the Company to publish an announcement of its results for any year or half-year under the GEM Listing Rules, or quarterly or any other interim period (whether or not required under the GEM Listing Rules), and ending on the date of the results announcement. The period during which no Option may be granted will cover any period of delay in the publication of a results announcement. The Board may not grant any Option to a Director during the periods or times in which Directors are prohibited from dealing in Shares pursuant to the Model Code for Securities Transactions by Directors of Listed Companies prescribed by the GEM Listing Rules or any other code or securities dealing restrictions adopted by the Company.
- 5.3 An Offer shall be made in writing (and unless so made shall be invalid) in such form as the Directors may from time to time determine, specifying the number of Shares in the Option and the Option Period in respect of which the Offer is made and further requiring the Selected Participant to undertake to hold the Option on the terms on which it is to be granted and to be bound by the provisions of this Scheme. Such Offer shall be personal to the Selected Participant concerned and shall not be transferable. It shall remain open for acceptance by the Selected Participant concerned (and by no other person, including his/her Personal Representative(s)) for a period of twenty-eight (28) days from the Offer Date provided that no such Offer shall be open for acceptance after the expiry of the Scheme Period or after this Scheme has been terminated or after the person to whom the Offer is made has ceased to be an Eligible Person.
- 5.4 An Offer shall state, in addition to the matters specified in Clause 5.3, the following:
  - (a) the name, address and position, if any, of the Selected Participant;
  - (b) the number of Shares in respect of which the Offer is made and the Subscription Price for such Shares;
  - (c) the Option Period in respect of which the Offer is made or, as the case may be, the Option Period in respect of separate parcels of Shares comprised in the Offer;
  - (d) the last date by which the Offer must be accepted;

- (e) the procedure for acceptance;
- (f) such other terms and conditions of the Offer as may be imposed by the Directors and which are not inconsistent with this Scheme; and
- (g) a statement requiring the Selected Participant to undertake to hold the Option on the terms on which it is to be granted and to be bound by the provisions of this Scheme including, without limitation, the conditions specified in Clause 7.1 and Clauses 17.8 to 17.10, inclusive.
- 5.5 An Offer shall be deemed to have been accepted and to have taken effect when the duplicate letter comprising acceptance of the Offer duly signed by the Selected Participant, together with a payment of HK\$1.00 by way of consideration for the grant thereof, is received by the Company within twenty-eight (28) days from the Offer Date. If such payment is not enclosed, the acceptance shall not be prejudiced thereby, but the acceptance shall create a promise by the Selected Participant to pay to the Company HK\$1.00 on demand. Such payment shall in no circumstances be refundable.
- 5.6 Any Offer may be accepted in respect of less than the number of Shares for which it is offered provided that it is accepted in respect of a board lot for dealing in Shares on the Stock Exchange or an integral multiple thereof and such number is clearly stated in the duplicate letter and accepted by the Selected Participant in the manner as specified in Clause 5.5.
- 5.7 Subject to Clause 10, upon an Offer being accepted by a Selected Participant in whole or in part in accordance with Clause 5.5 and, if appropriate, Clause 5.6, an Option in respect of the number of Shares in respect of which the Offer was so accepted will be deemed to have been granted by the Company to such Selected Participant on the date of such acceptance. To the extent that the Offer is not accepted within the time period set out in the Offer in the manner indicated in Clause 5.5 and, if appropriate, Clause 5.6, it will be deemed to have been irrevocably declined.
- 5.8 The Option Period of an Option shall not exceed ten (10) years from the Offer Date.
- 5.9 The Directors may, at their absolute discretion, fix any minimum period for which an Option must be held, any performance targets that must be achieved by the Grantee and any other conditions that must be fulfilled by the Grantee before the Option can be exercised upon the grant of an Option to a Selected Participant. If performance targets are imposed on a Grantee upon the grant of Options, the Board shall have regard to the purpose of the 2023 Share Option Scheme in assessing such performance targets, with reference to factors including but not limited to, as and when appropriate, sales performance (e.g. revenue), operating performance (e.g. profits, operation efficiency in terms of cost control), financial performance (e.g. profits, cash flow, earnings,

market capitalisation, return on equity) of the Group, corporate sustainability parameter (e.g. timeliness and accuracy in handling customer feedback, team work capabilities, adherence to corporate culture) and discipline and responsibility (e.g. punctuality, integrity, honesty or compliance with internal business procedures), the satisfaction of which shall be assessed and determined by the Board at its sole discretion. The Company shall utilise its internal assessment system to appraise and evaluate the performance targets applicable to each grant of Options on a case-by-case basis. The Company shall consider the past contributions of a Selected Participant with reference to the factors set out above and form an internal assessment as regards the future value that such Selected Participant may bring to the growth and development of the Group. The assessment shall involve the consideration and appraisal of the Selected Participant's expected contribution with reference to such Selected Participant's nature of duties (e.g. whether in a sale role, management role or a support role), position within the Group (e.g. whether overall Group level targets or specific performance indicators should be adopted) and other features including geographical location, corporate culture and business strategy focus. Specific weightings shall be given to the factors above in order to provide a fair and objective appraisal of the Selected Participants before the grant of Options, such that the grants will be on a fair and reasonable basis and in the interest of the Company and its Shareholders as a whole. The management shall propose the performance targets of each Selected Participant, if any, to the Board or the remuneration committee of the Company for consideration, who shall then assess the reasonableness and suitability of such performance targets.

- 5.10 The Vesting Period for any Option granted to any Grantee shall not be less than twelve (12) months from the date of grant of such Option, unless a shorter Vesting Period under specific circumstances as set out in this Scheme.
- 5.11 The Board or the remuneration committee of the Company shall have the authority to determine a shorter Vesting Period in its sole discretion under each of the following circumstances:
  - (a) grants of "make-whole" Options to a Selected Participant who newly joined the Group to replace the share options he/she forfeited when leaving the previous employers;
  - (b) grants to a Selected Participant whose employment is terminated due to disability or occurrence of any out of control event;
  - (c) grants that are made in batches during a year for administrative and compliance reasons, which may include Options that should have been granted earlier but had to wait for a subsequent batch, in such cases, the Vesting Date may be adjusted to take account of the time from which the Options would have been granted if not for such administrative or compliance requirements;

- (d) grants of Options with a mixed or accelerated vesting schedule such that the Options may vest evenly over a period of twelve (12) months; and
- (e) grants of Options with a total vesting and holding period of more than twelve (12) months.
- 5.12 The trustee holding unvested Shares of this Scheme, whether directly or indirectly, shall abstain from voting on matters that require Shareholders' approval under the GEM Listing Rules, unless otherwise required by law to vote in accordance with the beneficial owner's direction and such a direction is given.
- 5.13 In certain circumstances, it may be regarded as inequitable for any Shares under the Options to be vested or retained, as applicable. Such Options are therefore subject to Clawback, including but not limited to circumstances where there has been a material misstatement or omission in the financial statements of the Group made by a Selected Participant or if the relevant Selected Participant has engaged in serious negligence, fraud or misconduct, breached the Company's policies, rules or regulations or other circumstances. Notwithstanding anything else in this Scheme, any Shares under the Options may be subject to Clawback pursuant to the Company's Clawback mechanism, as amended from time to time.

#### 6. SUBSCRIPTION PRICE

The Subscription Price in respect of any Option shall, subject to any adjustments made pursuant to Clause 11, be such price as the Directors at their absolute discretion shall determine, provided that it must be no less than the higher of:

- (a) the closing price of Shares as stated in the Stock Exchange's daily quotations sheet on the Offer Date (which must be a Business Day);
- (b) the average closing price of Shares as stated in the Stock Exchange's daily quotations sheet for the five (5) Business Days immediately preceding the Offer Date; and
- (c) the par value of the Shares on the Offer Date.

#### 7. EXERCISE OF OPTIONS

- 7.1 An Option shall be personal to the Grantee and shall not be transferable or assignable and no Grantee shall in any way sell, transfer, charge, mortgage, encumber or otherwise dispose of or create any interest (whether legal or beneficial) whatsoever in favour of any third party over or in relation to any Option or enter into any agreement to do so. Any breach of the foregoing by a Grantee shall entitle the Company to cancel any Option granted to such Grantee to the extent not already exercised without incurring any liability on the part of the Company. The Stock Exchange may consider granting a waiver to allow a transfer any Options to a vehicle (such as a trust or a private company) for the benefit of the Grantee and any family members of such Grantee including but not limited to for estate planning and/or tax planning purposes that would continue to meet the purpose of this Scheme and comply with other requirements of Chapter 23 of the GEM Listing Rules. In the event of any such transfer, the Company shall disclose the beneficiaries of the trust or the ultimate beneficial owners of the transferee vehicle.
- 7.2 Subject to Clause 3.4 and Clause 17.8 and the fulfillment of all terms and conditions set out in the Offer, an Option shall be exercisable in whole or in part in the circumstances and in the manner as set out in Clauses 7.3 and 7.4 by giving notice in writing to the Company stating that the Option is thereby exercised and the number of Shares in respect of which it is so exercised (which, except where the number of Shares in respect of which the Option remains unexercised is less than one board lot or where the Option is exercised in full, must be for a board lot for dealings in Shares on the Stock Exchange or an integral multiple thereof). Each such notice must be accompanied by payment for the full amount of the Subscription Price for the Shares in respect of which the notice is given. Within twenty-eight (28) days (seven (7) days in the case of exercise pursuant to Clause 7.3(c)) after receipt of the notice and remittance and, where appropriate, receipt of the Auditors' or the Company's independent financial adviser's certificate pursuant to Clause 11, the Company shall accordingly allot and issue the relevant number of Shares to the Grantee (or, in the event of an exercise of Option by a Personal Representative pursuant to Clause 7.3(a), to the estate of the Grantee) credited as fully paid and issue to the Grantee (or his/her estate in the event of an exercise by his/her Personal Representative as aforesaid) a share certificate for the Shares so allotted and issued. When the Options are exercised only in part, the balance shall remain exercisable on the same terms as originally applied to the whole of the Options granted.

- 7.3 Subject as hereinafter provided, an Option may (and may only) be exercised by the Grantee at any time or times during the Option Period provided that:
  - (a) in the event of the Grantee ceases to be an Eligible Person thereof by reason of his/her death, permanent disability or retirement in accordance with his/her contract of employment and the termination of his/her employment is not on one or more of the grounds specified in Clause 8.1(c) before exercising the Option in full the Grantee or, as appropriate, his/her Personal Representative(s) may exercise the Option (to the extent not already exercised) in whole or in part in accordance with the provisions of Clause 7.2 within a period of twelve (12) months following the date of cessation of employment which date shall be the last day on which the Grantee was at work with the member of the Group whether salary is paid in lieu of notice or not, or such longer period as the Directors may determine, and any such Options not so exercised shall lapse and determine at the end of such period;
  - (b) in the event that the Grantee ceases to be an Eligible Person thereof by reason other than his/her death, permanent disability or retirement in accordance with his/her contract of employment or the termination of his/her employment is on one or more of the grounds specified in Clause 8.1(c) before exercising the Option in full, the Option (to the extent not already exercised) shall lapse on the date of cessation or termination and not be exercisable unless the Directors otherwise determine, in which event the Option (to the extent not already exercised) shall be exercisable in whole or in part within such period as the Directors may in its absolute discretion determine following the date of such cessation or termination. The date of cessation or termination as aforesaid shall be the last day on which the Grantee was actually at work with the member of the Group whether salary is paid in lieu of notice or not;
  - (c) if a general offer, whether by way of take-over offer, share re-purchase offer, or scheme of arrangement or otherwise in like manner is made to all Shareholders, and such offer becomes or is declared unconditional (in the case of take-over offer) or is approved by the requisite majorities at the relevant meeting of the Shareholders (in the case of a scheme of arrangement), the Directors shall as soon as practicable notify the Grantee (or his/her Personal Representative(s)) accordingly. The Grantee (or his/her Personal Representative(s)) shall be entitled to exercise his/her Option (to the extent not already exercised) either to its full extent or to the extent specified in the Grantee's notice to the Company at any time (in the case of takeover offer) within one (1) month after the date on which the offer becomes or is declared unconditional or (in the case of a scheme of arrangement) prior to such time and date as shall be notified by the Directors. To the extent that any Option has not been so exercised, it shall upon the expiry of such period lapse and determine;

- (d) in the event of a notice is given by the Company to its Shareholders to convene a general meeting for the purposes of considering, and if thought fit, approving a resolution to voluntarily wind-up the Company, the Company shall on the same date as or soon after it despatches such notice to each Shareholder of the Company give notice thereof to all Grantees and thereupon, each Grantee (or his/ her Personal Representative(s)) shall be entitled to exercise his/her Option (to the extent not already exercised) either to its full extent or to the extent specified in such notice in accordance with the provisions of Clause 7.2 at any time no later than two (2) Business Days prior to the proposed general meeting of the Company by giving notice to the Company in accordance with the provisions of Clause 7.2, accompanied by a remittance of the Subscription Price for the Shares in respect of which the notice is given. The Company shall as soon as possible and, in any event, no later than the Business Day immediately prior to the date of the proposed general meeting referred to above, allot the relevant Shares to the Grantee credited as fully paid; and
- if a compromise or arrangement between the Company and its members or (e) creditors is proposed for the purposes of or in connection with a scheme for the reconstruction of the Company or its amalgamation with any other company or companies, the Company shall give notice thereof to all Grantees (together with a notice of the existence of the provisions of this Clause 7.3(e)) on the same date as it despatches to each member or creditor of the Company a notice summoning the meeting to consider such a compromise or arrangement, and thereupon each Grantee (or his/her Personal Representative(s)) shall be entitled, at any time not later than (i) the Option Period, (ii) the period of two (2) months from the date of such notice, or (iii) the date on which such compromise or arrangement is sanctioned by the court, whichever is the earlier, to exercise his/her Options in whole or in part. The Company shall as soon as possible and in any event no later than one (1) Business Day prior to the date of such general meeting, allot and issue such number of Shares to the Grantees which fall to be issued on such exercise credited as fully paid. Subject thereto, all Options then outstanding shall lapse and determine upon such compromise or arrangement becoming effective.

7.4 The Shares to be allotted and issued upon the exercise of an Option will be subject to all the provisions of the memorandum of association of the Company and the Articles for the time being in force and will rank pari passu in all respects with the then existing fully paid Shares in issue on the date of allotment or, if that date falls on a day when the register of members of the Company is closed, the first day of the re-opening of the register of members (the "Allotment Date") and accordingly will entitle the holders thereof to participate in all dividends or other distributions paid or made on or after the Allotment Date or, if that date falls on a day when the register of members of the Company is closed, the first day of the re-opening of the register of members other than any dividend or other distribution previously declared or recommended or resolved to be paid or made if the record date therefore shall be before the Allotment Date. A Share allotted and issued upon the exercise of an Option shall not carry voting rights until the name of the Grantee has been duly entered on the register of members of the Company as the holder thereof.

#### 8. EARLY TERMINATION OF OPTION PERIOD

- 8.1 The Option Period in respect of any Option shall automatically terminate and that Option (to the extent not already exercised) shall lapse on the earliest of:
  - (a) the expiry of the Option Period;
  - (b) the expiry of any of the periods referred to in Clause 7.3;
  - (c) the date on which the Grantee ceases to be an employee by reason of voluntary resignation or dismissal or upon expiration of his/her term of directorship (unless immediately renewed upon expiration), or by a termination of his/her employment on the grounds that he/she has been guilty of persistent or serious misconduct, or has committed any act of bankruptcy or has become insolvent or has made any arrangement or composition with his/her creditors generally, or has been convicted of any criminal offence (other than an offence which in the opinion of the Directors does not bring the Grantee or the Group into disrepute or otherwise does not compromise the integrity or involve dishonesty of the Grantee) or any other grounds on which the relevant member of the Group would be entitled to terminate the Grantee's employment or directorship pursuant to any applicable laws. Transfer of employment or engagement or relationship from one member of the Group to another member of the Group shall not be considered as a cessation of employment, engagement or relationship;

- (d) subject to Clause 7.3(d), the date of the commencement of the winding-up of the Company; and
- (e) the date on which the Directors shall exercise the Company's right to cancel the Option by reason of breach of Clause 7.1 by the Grantee in respect of that or any other Option.
- 8.2 A resolution of the Board to the effect that the employment of a Grantee has or has not been terminated on one or more of the grounds specified in Clause 8.1(c) shall be conclusive.

## 9. MAXIMUM NUMBER OF SHARES AVAILABLE FOR SUBSCRIPTION

- 9.1 Subject to Clause 9.2 and Clause 11:
  - (a) The maximum number of the Shares which may be issued in respect of all Options and awards to be granted under all Share Schemes of the Company (including under this Scheme and any other Share Scheme(s)) shall not in aggregate exceed 10% of the Shares in issue as at the Adoption Date ("Scheme Mandate Limit").
  - (b) The Company may seek approval by the Shareholders in general meeting for refreshing the Scheme Mandate Limit under this Scheme after three (3) years from the Adoption Date or the date of the Shareholders' approval for the last refreshment (as the case may be).
  - (c) Any refreshment within any three-year period must be approved by Shareholders subject to the following provisions:
    - (i) any controlling Shareholders (or if there is no controlling Shareholder, the Directors (excluding independent non-executive Directors) and chief executive of the Company and their respective associates) and their associates must abstain from voting in favour of the relevant resolution at the general meeting; and
    - (ii) the Company must comply with the requirements under rules 17.47(6) and 17.47(7) of the GEM Listing Rules, Clause 10.4 to Clause 10.6 or such other provisions as required under Rule 23.03C of the GEM Listing Rules.

The requirements under this Clause 9.1(c) do not apply if the refreshment is made immediately after an issue of securities by the Company to its Shareholders on a pro rata basis as set out in rule 17.41(1) of the GEM Listing Rules such that the unused part of this Scheme Mandate Limit (as a percentage of the relevant class of Shares in issue) upon refreshment is the same as the unused part of the Scheme Mandate Limit immediately before the issue of securities, rounded to the nearest whole share.

- (d) The total number of Shares which may be issued in respect of all options and awards to be granted under all Share Scheme(s) (including this Scheme) under the Scheme Mandate Limit as refreshed shall not exceed 10% of the relevant class of Shares in issue as at the date of approval of the refreshed Scheme Mandate Limit by Shareholders in general meeting. The Company must send a circular to the Shareholders containing the number of Options that were already granted under the existing Scheme Mandate Limit, and the reason for the refreshment.
- (e) The Company may seek separate approval by the Shareholders in general meeting for granting Options beyond the Scheme Mandate Limit provided the Options in excess of the Scheme Mandate Limit are granted only to Eligible Persons specifically identified by the Company before such approval is sought. The Company must send a circular to the Shareholders containing the name of each specified Eligible Person who may be granted such Options, the number and terms of the Options to be granted to each Eligible Persons, the purpose of granting Options to the specified Eligible Persons with an explanation as to how the terms of the Options serve such purpose. The number and terms of Options to be granted to such Eligible Persons must be fixed before Shareholders' approval. In respect of any Options to be granted, the date of the Board meeting for proposing such grant should be taken as the Offer Date for the purpose of calculating the Subscription Price under Clause 6.
- 9.2 If the Company conducts a share consolidation or subdivision after the Scheme Mandate Limit has been approved by the Shareholders in general meeting, the maximum number of Shares that may be issued upon exercise of all options to be granted under this Scheme and any other Share Scheme(s) under the Scheme Mandate Limit as a percentage of the total number of issued Shares at the date immediately before and after such consolidation or subdivision shall be the same, rounded to the nearest whole Share.

- 9.3 Where any grant of Options and awards to an Eligible Person would result in the Shares issued and to be issued in respect of all Options and awards granted and to be granted to such Eligible Person (excluding options and awards that have been lapsed in accordance with this Scheme and any other Share Scheme(s)) in the 12-month period up to and including the date of such grant representing in aggregate over 1% of the Shares in issue (the "1% Individual Limit"), such grant must be separately approved by the Shareholders in general meeting with such Eligible Person and his/her close associates (or his/her associates if the Eligible Person is a connected person) abstaining from voting. The Company must send a circular to the Shareholders and the circular must disclose the identity of the Eligible Person, the number and terms of the Options and awards to be granted (and those previously granted to such Eligible Person in the 12-month period), the purpose of granting Options and awards to the Eligible Person and an explanation as to how the terms of the Options serve such purpose. The number and terms (including the Subscription Price) of Options to be granted to such Eligible Person must be fixed before Shareholders' approval. In respect of any Options and awards to be granted, the date of board meeting for proposing such grant should be taken as the Offer Date for the purpose of calculating the Subscription Price under Clause 6.
- 9.4 Subject to Clauses 9.1, 9.2 and 9.3, the maximum number of Shares subject to Options and to this Scheme may be adjusted in such manner as the Auditors or the independent financial adviser to the Company shall certify in writing to the Directors to be in their opinion fair and reasonable, in accordance with Clause 11, in the event of any alteration in the capital structure of the Company whether by way of capitalisation of profits or reserves, rights issue, consolidation, sub-division or reduction of the share capital of the Company otherwise howsoever.

#### 10 GRANT OF OPTIONS TO CONNECTED PERSONS AND VOTING PROCEDURES

10.1 Any grant of Options to a director, chief executive or substantial shareholder of the Company or any of their respective associates must be approved by the independent non-executive Directors (excluding any independent non-executive Director who is a Grantee) and shall comply with the relevant provisions of Chapter 23 of the GEM Listing Rules. 10.2 Where any grant of Options to a substantial shareholder or an independent non-executive Director or any of their respective associates would result in the Shares issued and to be issued in respect of all Options under this Scheme and all options, awards and securities granted under other Share Scheme(s) (including Options exercised, cancelled and outstanding but excluding those options, awards and securities lapsed in accordance with the terms of such schemes) to that person in the 12-month period up to and including the date of such grant representing in aggregate over 0.1% of the total number of Shares in issue, such grant of Options must be approved by the Shareholders in general meeting in the manner set out in rule 23.04(4) of the GEM Listing Rules. The Grantee, his/her associates and all core connected persons of the Company must abstain from voting in favour at such general meeting. The Company must comply with the requirements under rule 23.04 of the GEM Listing Rules.

The Company must send a circular sent to the Shareholders, explaining the proposed grant, which shall contain:

- (a) details of the number and terms (including the Subscription Price) of the Options to be granted to each Selected Participant, which must be fixed before the Shareholders' meeting, and the date of meeting of the Board for proposing such further grant is to be taken as the date of grant for the purpose of calculating the Subscription Price under rule 23.03E of the GEM Listing Rules. The description of the terms of the Options must include the information required under rules 23.03(5) to 23.03(10) and rule 23.03(19) of the GEM Listing Rules;
- (b) the views of the independent non-executive Directors (excluding any independent non-executive Director who is the Grantee) as to whether the terms of the grant are fair and reasonable and whether such grant is in the interests of the Company and the Shareholders as a whole, and their recommendation to the independent Shareholders as to voting;
- (c) the information required under rule 23.02(2)(c) of the GEM Listing Rules; and
- (d) the information required under rule 2.28 of the GEM Listing Rules.

- 10.3 Any vote taken at the general meeting must be taken on a poll. The Company shall announce the results of the poll (including (i) the total number of Shares entitling the holder to attend and vote for or against the resolution at the general meeting, (ii) the total number of Shares entitling the holder to attend and vote only against the resolution at the general meeting, (iii) the number of Shares represented by votes for and against the relevant resolution) by way of an announcement which is published in accordance with rules 16.17 to 16.19 of the GEM Listing Rules as soon as possible, but in any event not later than the time that is 30 minutes before the earlier of the commencement of the morning trading session or any pre-opening session on the Business Day following the meeting. The Company shall appoint its auditors, share registrar or external accountants who are qualified to serve as auditors for the Company as scrutineer for the vote-taking and state the identity of the scrutineer in the announcement. The Company shall confirm in the announcement whether or not any parties that have stated their intention in the circular to vote against the relevant resolution or to abstain have done so at the general meeting.
- 10.4 Parties that are required to abstain from voting in favour at the general meeting referred to in Clause 9.1(c) and Clause 10.2 may vote against the resolution at the general meeting of the Company provided that their intention to do so has been stated in the circular to Shareholders. Any such party may change his/her mind as to whether to abstain or vote against the resolution, in which case the Company must, if it becomes aware of the change before the date of the general meeting, immediately despatch a circular to the Shareholders or publish an announcement notifying the Shareholders of the change and, if known, the reason for such change. Where the circular is despatched or the announcement is published less than ten (10) Business Days before the date originally scheduled for the general meeting, the general meeting must be adjourned before considering the relevant resolution to a date that is at least ten (10) Business Days from the date of despatch or publication by the chairman or, if that is not permitted by the Articles, by resolution to that effect.
- 10.5 Where a general meeting is required to be adjourned by resolution, all Shareholders are permitted to vote on that resolution. Any Shareholders who would have been required to abstain from voting on any resolution that was to be proposed shall vote in favour of the resolution to adjourn the general meeting.
- 10.6 The Company must have an appropriate procedure in place to record that any parties that must abstain or have stated their intention to vote against the relevant resolution in the circular or announcement have done so at the general meeting.

#### 11 REORGANISATION OF CAPITAL STRUCTURE

- 11.1 In the event of any alteration in the capital structure of the Company whilst any Option remains exercisable or this Scheme remains in effect, and such event arises from a capitalisation of profits or reserves, rights issue, consolidation, sub-division or reduction of the share capital of the Company, then, in any such case, the Company shall instruct the Auditors or the independent financial adviser to the Company to certify in writing the adjustment, if any, that ought in their opinion to be fair and reasonable to be made either generally or as regards any particular Grantee, to:
  - (a) the number of Shares to which this Scheme or any Option(s) relates (insofar as it is/they are unexercised); and/or
  - (b) the Subscription Price of any unexercised Option; and/or
  - (c) the number of Shares in respect of which any further Options may be granted within the 1% Individual Limit and/or the Scheme Mandate Limit.

and an adjustment as so certified by the Auditors or the independent financial adviser to the Company shall be made, provided that:

- (i) the aggregate Subscription Price payable by a Grantee on the full exercise of any Option shall remain as nearly as possible the same (but shall not be greater than) as it was before such event;
- (ii) no such adjustment shall be made the effect of which would be to enable a Share to be issued at less than its nominal value;
- (iii) no such adjustment shall be made the effect of which would increase the proportion of the equity capital of the Company, rounded to the nearest whole Share, for which any Grantee would have been entitled to subscribe had he/she exercised all the Options held by him/her immediately prior to such adjustment;
- (iv) the issue of Shares or securities of the Company as consideration in a transaction shall not be regarded as a circumstance requiring any such adjustment; and

(v) for the avoidance of doubt, any adjustment shall be made in compliance with the GEM Listing Rules or other relevant guidance as the Stock Exchange may from time to time issue.

In respect of any such adjustments, other than any made on a capitalisation issue, the Auditors or the independent financial adviser to the Company must confirm to the Directors in writing that such adjustments satisfy the requirements set out in the Note to rule 23.03(13) of GEM Listing Rules.

- 11.2 If there has been any alteration in the capital structure of the Company as referred to in Clause 11.1, the Company shall, upon receipt of a notice from a Grantee in accordance with Clause 7.2, inform the Grantee of such alteration and shall either inform the Grantee of the adjustment to be made in accordance with the certificate of the Auditors or the independent financial adviser to the Company obtained by the Company for such purpose or, if no such certificate has yet been obtained, inform the Grantee of such fact and instruct the Auditors or the independent financial adviser to the Company as soon as practicable thereafter to issue a certificate in that regard in accordance with Clause 11.1.
- 11.3 In giving any certificate under this Clause 11 the Auditors or independent financial adviser to the Company shall be deemed to be acting as experts and not as arbitrators and their certificate shall, in the absence of manifest error, be final and binding on the Company and all persons who may be affected thereby.

## 12. SHARE CAPITAL

The exercise of any Option shall be subject to the Shareholders in general meeting approving any necessary increase in the authorised share capital of the Company. Subject thereto, the Directors shall make available sufficient authorised but unissued share capital of the Company to allot and issue the Shares on the exercise of Options.

## 13. DISPUTES

Any dispute arising in connection with the number of Shares the subject of an Option, any of the matters referred to in Clause 9.4, or any adjustment under Clause 11.1 shall be referred to the decision of the Auditors or the independent financial adviser to the Company who shall act as experts and not as arbitrators and whose decision shall, in the absence of manifest error, be final and binding on all persons who may be affected thereby.

#### 14. ALTERATION OF THIS SCHEME

- 14.1 Any change to the terms of the Options granted to a Selected Participant must be approved by the Board, the remuneration committee of the Company, the independent non-executive Directors and/or the Shareholders (as the case may be) if the initial grant of the Option was approved by the Board, the remuneration committee of the Company, the independent non-executive Directors and/or the Shareholders (as the case may be), except where the alterations take effect automatically under the existing terms of this Scheme.
- 14.2 Any alterations to the terms and conditions of this Scheme which are of a material nature or any alterations to the provisions relating to the matters set out in Rule 23.03 of the GEM Listing Rules to the advantage of Selected Participants must be approved by the Shareholders in general meeting.
- 14.3 Any change to the authority of the Directors or the scheme administrators in relation to any alteration to the terms of this Scheme shall be approved by the Shareholders in a general meeting.
- 14.4 The amended terms of this Scheme or the Options must still comply with the applicable requirements under the GEM Listing Rules.

#### 15. TERMINATION

- 15.1 The Company may by resolution in general meeting at any time terminate the operation of this Scheme and in such event no further Options will be offered or granted but in all other respects the provisions of this Scheme shall remain in force to the extent necessary to give effect to the exercise of any outstanding Options granted prior thereto or otherwise as may be required in accordance with the provisions of this Scheme and outstanding Options granted prior to such termination shall continue to be valid and exercisable in accordance with the provisions of this Scheme.
- 15.2 Details of the Options granted, including Options exercised or outstanding, under this Scheme and (if applicable) Options that become void or non-exercisable as a result of the termination must be disclosed in the circular to Shareholders seeking approval of the first new scheme to be established or refreshment of the Scheme Mandate Limit under any existing Share Scheme(s) after such termination.

#### 16. CANCELLATION OF OPTION GRANTED

- 16.1 The Directors may effect the cancellation of any Options granted but not exercised on such terms as may be agreed with the relevant Grantee, as the Directors may in their absolute discretion see fit and in a manner that complies with all applicable legal requirements for cancellation. Where the Company cancels any Options granted and make a new grant of Options to the same Grantee, such new grant of Options may only be made with the available unissued Options (excluding the cancelled Options) under this Scheme within the Scheme Mandate Limit.
- 16.2 Options cancelled in accordance with the terms of this Scheme will be regarded as utilised for the purpose of calculating the Scheme Mandate Limit. Options lapsed in accordance with the terms of this Scheme will not be regarded as utilised for the purpose of calculating the Scheme Mandate Limit.

#### 17. MISCELLANEOUS

- 17.1 This Scheme shall not form part of any contract of employment between the Company or any Subsidiary and any employee and the rights and obligations of any employee under the terms of his/her office or employment shall not be affected by his/her participation in this Scheme or any right which he/she may have to participate in it and this Scheme shall afford such employee no additional rights to compensation or damages in consequence of the termination of such office or employment for any reason.
- 17.2 This Scheme shall not confer on any person any legal or equitable rights (other than those constituting the Options themselves) against the Company directly or indirectly or give rise to any cause of action at law or in equity against the Company.
- 17.3 The Company shall bear the costs of establishing and administering this Scheme, including any costs of the Auditors or the independent financial adviser to the Company in relation to the preparation of any certificate by them or providing any other service in relation to this Scheme.
- 17.4 A Grantee shall be entitled to receive copies of all notices and other documents sent by the Company to Shareholders at the same time or within a reasonable time of any such notices or documents being sent to Shareholders and for such purpose, publication by the Company of the notices and other documents on the websites of the Company and the Stock Exchange shall be deemed to be due and sufficient delivery of the same to the Grantee.

- 17.5 Any notice or other communication between the Company and a Selected Participant may be given by sending the same by prepaid post or by personal delivery to, in the case of the Company, its principal place of business in Hong Kong and, in the case of the Selected Participant, his/her address in Hong Kong or the People's Republic of China as notified to the Company from time to time or, if none or incorrect or out of date, his/her last place of employment with the Company or the Company's principal place of business in Hong Kong from time to time.
- 17.6 Any notice or other communication if sent by the Selected Participant shall be irrevocable and shall not be effective until actually received by the Company.
- 17.7 Any notice or other communication if sent to the Selected Participant shall be deemed to be given or made:
  - (a) one (1) day after the date of posting, if sent by mail; and
  - (b) when delivered, if delivered by hand.
- 17.8 A Grantee shall, before accepting an Offer or exercising his/her Option, obtain all necessary consents that may be required to enable him/her to accept the Offer or to exercise the Option and the Company to allot and issue to him/her in accordance with the provisions of this Scheme the Shares falling to be allotted and issued upon the exercise of his/her Option. By accepting an Offer or exercising his/her Option, the Grantee thereof is deemed to have represented to the Company that he/she has obtained all such consents. Compliance with this Clause 17.8 shall be condition precedent to an acceptance of an Offer by a Grantee and an exercise by a Grantee of his/her Options. A Grantee shall indemnify the Company fully against all claims, demands, liabilities, actions, proceedings, fees, costs and expenses which the Company may suffer or incur (whether alone or jointly with other party or parties) for or in respect of any failure on the part of the Grantee to obtain any necessary consent or to pay tax or other liabilities referred therein.
- 17.9 A Grantee shall ascertain and pay all tax and discharge all other liabilities to which he/she may become subject as a result of his/her participation in this Scheme, the acceptance of an Offer or the exercise of any Option.
- 17.10 By accepting an Offer, an Eligible Person shall be deemed irrevocably to have waived any entitlement, by way of compensation for loss of office or otherwise howsoever, to any sum or other benefit to compensate him/her for loss of any rights or benefits under this Scheme.

- 17.11 In no circumstances shall a person ceasing to be an Eligible Person for any reason be entitled to any compensation for or in respect of any consequent diminution or extinction of his/her/its rights or benefits (actual or prospective) under any Options then held by him/her or otherwise in connection with this Scheme and by accepting an Option under this Scheme, such person shall be deemed to have irrevocably waived any such entitlement (to the extent there is any).
- 17.12 The Company shall provide a summary of the terms of this Scheme to all Eligible Persons on joining this Scheme (and a copy of this Scheme document to any Eligible Person who requests such a copy). The Company shall provide to all Eligible Persons all details relating to changes in the terms of this Scheme during the life of this Scheme immediately upon such changes taking effect.
- 17.13 This Scheme and all Options granted hereunder shall be governed by and construed in accordance with the laws of Hong Kong.
- 17.14 This Scheme shall comply with the GEM Listing Rules as amended from time to time. In the event that there are differences between the terms of this Scheme and the GEM Listing Rules, the GEM Listing Rules shall prevail.

## NOTICE OF EGM



# LEGENDARY EDUCATION GROUP LIMITED

# 傳承教育集團有限公司

(formerly known as Legendary Group Limited 創天傳承集團有限公司)
(incorporated in the Cayman Islands with limited liability)
(Stock Code: 8195)

## NOTICE OF EXTRAORDINARY GENERAL MEETING

**NOTICE IS HEREBY GIVEN THAT** the extraordinary general meeting (the "**EGM**") of Legendary Education Group Limited (the "**Company**") will be held at 11:00 a.m. on Friday, 1 September 2023 at 5/F, World Interests Building, 8 Tsun Yip Lane, Kwun Tong, Kowloon, Hong Kong for the following purpose of considering and, if thought fit, passing with or without amendment, the following resolutions as ordinary resolutions:

## **ORDINARY RESOLUTIONS**

## "THAT

- 1. the share option scheme adopted by the Company on 25 September 2014 be and is hereby terminated (save with respect to any outstanding, issued and unexercised options thereof) with effect from the approval of the Shareholders at the EGM subject to and conditional upon the 2023 Share Option Scheme (as defined below) becoming unconditional and effective; and
- 2. subject to and conditional upon the passing of the resolution in paragraph (1) hereinabove and the GEM Listing Committee of The Stock Exchange of Hong Kong Limited ("Stock Exchange") granting the approval for the listing of, and permission to deal in, the shares in the capital of the Company ("Shares") which may fall to be issued and allotted pursuant to the exercise of any options that may be granted under the share option scheme of the Company (the rules of which are contained in the document marked "A" produced to the EGM and signed by the chairman of the EGM for the purpose of identification) ("2023 Share Option Scheme"), the 2023 Share Option Scheme be and is hereby approved and adopted and the directors of the Company be and are hereby authorised to do all such acts and to enter into all such transactions, arrangements and agreements as may be necessary or expedient in order to give full effect to the 2023 Share Option Scheme including without limitation:
  - (a) to administer the 2023 Share Option Scheme under which options will be granted to Eligible Participants (as defined in the 2023 Share Option Scheme) to subscribe for Shares;

## NOTICE OF EGM

- (b) to modify and/or amend the 2023 Share Option Scheme from time to time provided that such modification and/or amendment is effected in accordance with the provisions of the 2023 Share Option Scheme relating to modification and/or amendment;
- (c) to issue and allot from time to time such number of Shares as may be required to be issued pursuant to the exercise of the options under the 2023 Share Option Scheme and subject to the Rules Governing the Listing of Securities on GEM of the Stock Exchange;
- (d) to make application at the appropriate time or times to the Stock Exchange for the listing of, and permission to deal in any Shares which may thereafter from time to time be issued and allotted pursuant to the exercise of the options under the 2023 Share Option Scheme; and
- (e) to consent, if they deem fit and expedient, to such conditions, modifications and/or variations as may be required or imposed by the relevant authorities in relation to the 2023 Share Option Scheme."

By order of the Board

Legendary Education Group Limited

Yuen Yu Sum

Chairman and Executive Director

Hong Kong, 16 August 2023

Registered Office:
Maples Corporate Services Limited
P.O. Box 309, Ugland House
Grand Cayman, KY1-1104
Cayman Islands

Head office and principal place of
business in Hong Kong:
5/F, World Interests Building
8 Tsun Yip Lane
Kwun Tong
Kowloon
Hong Kong

## NOTICE OF EGM

#### Notes:

- 1. A member entitled to attend and vote at the EGM is entitled to appoint one or more proxy to attend and, subject to the provisions of the Articles of Association, to vote on his/her behalf. A proxy need not be a member of the Company but must be present in person at the EGM to represent the member. If more than one proxy is so appointed, the appointment shall specify the number and class of shares in respect of which each such proxy is so appointed.
- 2. In order to be valid, the form of proxy and the power of attorney (if any), under which it is signed or a notarially certified copy thereof, must be lodged, at the office of the Company's branch share registrar and transfer office in Hong Kong, Union Registrars Limited at Suites 3301-04, 33/F., Two Chinachem Exchange Square, 338 King's Road, North Point, Hong Kong not less than 48 hours before the time appointed for the holding of the EGM or any adjourned meeting.
- 3. Completion and return of a form of proxy will not preclude members of the Company from attending and voting in person at the EGM or any adjournment thereof should they so wish and in such event, the form of proxy shall be deemed to be revoked.
- 4. Where there are joint registered holders of any Shares, any one of such persons may vote at the EGM, either personally or by proxy, in respect of such Shares as if he were solely entitled thereto; but if more than one of such joint holders be present at the EGM personally or by proxy, that one of the said person as present whose name stands first on the register in respect of such Share shall alone be entitled to vote in respect thereof.
- 5. For the purpose of determining the entitlement to attend and vote at the EGM, the register of members of the Company will be closed from Tuesday, 29 August 2023 to Friday, 1 September 2023, both days inclusive, during which period no transfer of Shares will be registered. In order to qualify to attend and vote at the EGM, all transfer documents accompanied by the relevant share certificates must be lodged with the Company's branch share registrar and transfer office in Hong Kong, Union Registrars Limited at Suites 3301-04, 33/F., Two Chinachem Exchange Square, 338 King's Road, North Point, Hong Kong, for registration not later than 4:00 p.m. on Monday, 28 August 2023.
- 6. As at the date of this notice, the Board comprises Mr. Yuen Yu Sum (Chairman) and Mr. Chan Lap Jin Kevin as executive Directors, Mr. Law Wing Chung and Dr. Tang Sing Hing Kenny as non-executive Directors and Mr. Chung Chin Kwan, Mr. Chan Kim Fai Eddie and Mr. Chung Kwok Pan as independent non-executive Directors.
- 7. If Typhoon Signal No. 8 or above, or a "black" rainstorm warning or "external conditions after super typhoons" announced by the Government of Hong Kong is/are in effect any time after 7:00 a.m. on the date of the EGM, the EGM will be postponed. The Company will publish an announcement on the website of the Company at www.legendaryedu.com and on the "Latest Company Announcements" page of the Stock Exchange's website at www.hkexnews.hk to notify shareholders of the Company of the date, time and place of the rescheduled meeting.